Author's response to reviews

Title: Identification of effective screening strategies for cardiovascular disease prevention in a developing country: using cardiovascular risk-estimation and risk-reduction tools for policy recommendations

Authors:

Sharmini Selvarajah (s.selvarajah@umcutrecht.nl)
Jamaiyah Haniff (jamaiyah@_crc.gov.my)
Gurpreet Kaur (dr.gurpreet@iku.moh.gov.my)
Tee Guat Hiong (ghiong88@yahoo.com)
Adam Bujang (adam@crc.gov.my)
Kee Chee Cheong (kee@imr.gov.my)
Michiel L Bots (m.l.bots@umcutrecht.nl)

Version: 3 Date: 18 October 2012

Author's response to reviews: see over
Dear Dr Timothy Shipley,

Thank you for considering the revised version of our manuscript entitled “Identification of effective screening strategies for cardiovascular disease prevention in a developing country: using cardiovascular risk-estimation and risk-reduction tools for policy recommendations” for publication in BMC Cardiovascular Disorders. We are grateful to the reviewers for pointing out some important clarifications and changes needed in the paper. We have thoughtfully taken into account these comments and have addressed the reviewers’ concerns point by point in the following pages.

We consider the comments very constructive and have been especially useful in restructuring the manuscript.

We hope all these changes fulfill the requirements to make this manuscript acceptable for publication in BMC Cardiovascular Disorders.

Looking forward to hearing from you soon.

Yours sincerely,

Sharmini Selvarajah